Author's response to reviews

Title: Non-European Union doctors in the National Health Service: why, when and how do they come to the United Kingdom of Great Britain and Northern Ireland?

Authors:

Jyothis T George (dritgeorge@yahoo.co.uk)
Kavitha S Rozario (kavitharoz@yahoo.com)
Jeffrin Anthony (jeffrinanthony@gmail.com)
Edward B Jude (edward.jude@tgh.nhs.uk)
Gerard A McKay (gerard.mckay@lanarkshire.scot.nhs.uk)

Version: 4 Date: 26 February 2007

Author's response to reviews: see over
Sir/Madam,

Thank you for considering our research work, looking at the factors driving migration of non-European doctors to the United Kingdom.

Our work shows that the primary reason for many non-European doctors to migrate to the UK is for training within the National Health Service (NHS) of UK. In a climate where international migration of health professionals is of international concern, we believe our work will add to the body of knowledge in this domain.

Working with the editor as well as the reviewers has been a great positive experience. On behalf of the authors, I would like to express sincere gratitude to their time and effort towards making this article better.

Formatting changes requested by the editorial team have been addressed (see below) and we have great pride in submitting this final revision to Human Resources in Health.

Thanking you,

Sincerely,

Dr. Jyothis Thomas George
Corresponding Author

Changes in current version:-
   a) Author names edited as advised
   b) ‘Et al’ in reference substituted by the full list of authors
   c) Referencing sequence of figures checked – results section edited.
   d) Author’s contributions – Edited as suggested.